\begin{table}%C.3
%\centering
\par
%==============================================================================
\caption{Ground-based near-infrared (${\it JHKLM}$) photometry found in
the literature. If there was more than one measurement we used both
the maximum and minimum datapoints. For the data from 2~MASS and
DENIS we made use of the catalogues found in VIZIER
({\tt http://vizier.u-strasbg.fr/viz-bin/VizieR}).\label{DATA:NIR}}
\begin{tabular}{lllllllllll}
\hline
\hline
No	&Name			&JD		&$I$	&$J$	&$H$	&$K$	&$L$	&$M$	&System		&Reference\\
	&			&2400000	&	&	&	&	&	&	&		&\\
\hline
1	&TW~Cam		&		&	&	&	&	&4.2	&3.5	&Mt Lemmon	&\citet{72bGehrz}\\
	&			&		&	&	&	&	&	&	&Arizona system	&\\
	&			&42445.45	&	&7.04	&6.36	&5.70	&4.40	&	&Tenerife	&\citet{85LloydEvans}\\
	&			&		&	&7.035	&6.364	&5.750	&	&	&2~MASS	&VIZIER\\
\par
2	&RV~Tau		&		&	&	&	&5.1	&3.3	&2.3	&Mt Lemmon	&\citet{72bGehrz}\\
	&			&		&	&	&	&	&	&	&Arizona system	&\\
	&			&42445.48	&	&6.39	&5.67	&5.03	&3.64	&	&Tenerife	&\citet{85LloydEvans}\\
	&			&		&	&6.183	&5.488	&4.777	&	&	&2~MASS	&VIZIER\\
\par
3	&IRAS~05208		&48637.35	&	&7.05	&6.24	&6.05	&5.64	&	&SAAO		&T.~Lloyd Evans\\
	&			&49313.47	&	&7.20	&6.38	&6.13	&5.64	&	&SAAO		&T.~Lloyd Evans\\
	&			&		&8.705	&6.408	&	&5.953	&	&	&DENIS		&VIZIER\\
	&			&		&	&7.044	&6.314	&6.063	&	&	&2~MASS	&VIZIER\\
\par
4	&DY~Ori		&43189.30	&	&8.10	&7.47	&7.02	&5.73	&	&SAAO		&\citet{85LloydEvans}\\
	&			&43158.44	&	&8.30	&7.40	&6.90	&5.80	&	&SAAO		&\citet{85LloydEvans}\\
	&			&		&	&8.073	&7.381	&6.983	&	&	&2~MASS	&VIZIER\\
\par
5	&CT~Ori		&43184.37	&	&8.57	&7.82	&8.00	&6.48	&	&SAAO		&\citet{85LloydEvans}\\
	&			&43158.38	&	&8.66	&7.92	&7.50	&6.09	&	&SAAO		&\citet{85LloydEvans}\\
	&			&		&	&8.404	&7.931	&7.607	&	&	&2~MASS	&VIZIER\\
\par
6	&SU~Gem		&		&	&	&	&	&4.1	&3.2	&Mt Lemmon	&\citet{72bGehrz}\\
	&			&		&	&	&	&	&	&	&Arizona system	&\\
	&			&43160.37	&	&8.36	&7.38	&6.47	&4.94	&	&SAAO		&\citet{85LloydEvans}\\
	&			&43188.32	&	&8.97	&8.38	&6.95	&4.96	&	&SAAO		&\citet{85LloydEvans}\\
	&			&		&	&10.167	&9.033	&7.584	&	&	&2~MASS	&VIZIER\\
\par
7	&UY~CMa		&46107.32	&	&9.35	&8.82	&8.22	&6.62	&	&SAAO		&\citet{85LloydEvans}\\
	&			&43159.36	&	&9.42	&8.46	&8.41	&6.87	&	&SAAO		&\citet{85LloydEvans}\\
	&			&		&9.845	&9.093	&	&8.038	&	&	&DENIS		&VIZIER\\
	&			&		&	&9.042	&8.597	&7.986	&	&	&2~MASS	&VIZIER\\
\par
8	&HD~44179		&		&	&6.59	&4.95	&3.19	&1.22	&	&ESO		&C.~Waelkens\\
	&			&		&8.981	&6.280	&& 3.807	&	&	&DENIS		&VIZIER\\
	&			&		&	&6.577	&5.145	&3.655	&	&	&2~MASS	&VIZIER\\
\par
9	&HD~46703		&		&	&7.805	&7.513	&7.435	&	&	&2MASS		&VIZIER\\
\par
10	&ST~Pup		&43159.44	&	&8.38	&8.03	&7.67	&6.42	&	&SAAO		&\citet{85LloydEvans}\\
	&			&		&9.292	&8.437	&	&7.839	&	&	&DENIS		&VIZIER\\
	&			&		&	&8.506	&8.250	&8.082	&	&	&2~MASS	&VIZIER\\
\par
11	&HD~52961		&		&	&6.36	&5.98	&5.54	&4.25	&3.97	&ESO		&C.~Waelkens\\
	&			&		&	&6.323	&5.978	&5.526	&	&	&2~MASS	&VIZIER\\
\par
12	&SAO~173329		&		&	&9.42	&8.96	&8.27	&6.49	&5.67	&ESO		&C.~Waelkens\\
	&			&		&	&9.375	&8.917	&8.308	&	&	&2~MASS	&VIZIER\\
\par
13	&U~Mon			&		&	&	&	&	&2.4	&1.5	&Kitt Peak NO	&\citet{70Gehrz}\\
	&			&		&	&	&	&	&	&	&Arizona system	&\\
	&			&		&	&	&	&3.7	&2.7	&1.7	&Mt Lemmon	&\citet{72bGehrz}\\
	&			&		&	&	&	&	&	&	&Arizona system	&\\
	&			&42684.65	&	&4.01	&3.65	&3.25	&2.40	&	&SAAO		&\citet{85LloydEvans}\\
	&			&43160.45	&	&4.88	&4.43	&4.08	&3.08	&	&SAAO		&\citet{85LloydEvans}\\
	&			&		&	&4.35	&3.93	&3.60	&2.30	&1.61	&ESO		&C.~Waelkens\\
	&			&		&	&4.925	&4.269	&4.042	&	&	&2~MASS	&VIZIER\\
\par
14	&AR~Pup		&43184.46	&	&6.12	&4.95	&3.67	&1.68	&	&SAAO		&\citet{85LloydEvans}\\
	&			&42737.58	&	&7.20	&5.98	&4.40	&2.08	&	&SAAO		&\citet{85LloydEvans}\\
	&			&		&8.937	&6.931	&	&3.682	&	&	&DENIS		&VIZIER\\
	&			&		&	&7.891	&6.824	&5.285	&	&	&2~MASS	&VIZIER\\
\par
15	&IRAS~08544		&		&	&5.65	&4.67	&3.51	&1.59	&	&SAAO		&T.~Lloyd Evans (mean)\\
	&			&		&9.043	&5.741	&	&3.813	&	&	&DENIS		&VIZIER\\
	&			&		&	&5.575	&4.743	&3.523	&	&	&2~MASS	&VIZIER\\
\par
	\hline
16	&IRAS~09060		&50263.20	&	&8.60	&7.63	&6.68	&5.16	&	&SAAO		&T.~Lloyd Evans\\
	&			&50511.38	&	&9.27	&8.13	&6.99	&5.35	&	&SAAO		&T.~Lloyd Evans\\
	&			&		&	&9.130	&8.060	&6.949	&	&	&2~MASS	&VIZIER\\
17	&IRAS~09144		&49125.31	&	&8.33	&7.12	&6.00	&4.23	&	&SAAO		&T.~Lloyd Evans\\
	&			&49475.28	&	&8.55	&7.35	&6.21	&4.31	&	&SAAO		&T.~Lloyd Evans\\
	&			&		&10.801	&8.305	&	&6.235	&	&	&DENIS		&VIZIER\\
	&			&		&	&8.372	&7.391	&6.369	&	&	&2~MASS	&VIZIER\\
\par
18	&IW~Car		&43184.48	&	&5.86	&5.18	&4.24	&2.41	&	&SAAO		&\citet{85LloydEvans}\\
	&			&43155.56	&	&5.94	&5.20	&4.31	&2.44	&	&SAAO		&\citet{85LloydEvans}\\
	&			&		&	&5.875	&5.151	&4.367	&	&	&2~MASS	&VIZIER	\\
\par
19	&IRAS~09400		&49122.29	&	&6.57	&5.58	&5.16	&4.33	&	&SAAO		&T.~Lloyd Evans\\
	&			&50511.40	&	&6.61	&5.59	&5.20	&4.35	&	&SAAO		&T.~Lloyd Evans\\
	&			&		&	&6.547	&5.626	&5.159	&	&	&2~MASS	&VIZIER\\
\par
20	&IRAS~09538		&49741.57	&	&9.18	&8.49	&7.61	&6.09	&	&SAAO		&T.~Lloyd Evans\\
	&			&49031.52	&	&9.62	&8.82	&8.02	&6.36	&	&SAAO		&T.~Lloyd Evans\\
	&			&		&10.530	&9.278	&	&7.703	&	&	&DENIS		&VIZIER\\
	&			&		&	&9.402	&8.574	&7.721	&	&	&2~MASS	&VIZIER\\
\par
21	&HR~4049		&		&	&4.73	&4.16	&3.13	&1.39	&0.91	&ESO		&C.~Waelkens\\
	&			&		&	&	&	&3.677	&	&	&DENIS		&VIZIER\\
	&			&		&	&4.921	&4.225	&3.208	&	&	&2~MASS	&VIZIER\\
\par
22	&IRAS~10174		&49474.35	&	&5.20	&4.07	&3.54	&2.87	&	&SAAO		&T.~Lloyd Evans\\
	&			&50262.22	&	&5.43	&4.29	&3.74	&3.04	&	&SAAO		&T.~Lloyd Evans\\
	&			&		&8.644	&5.583	&	&3.481	&	&	&DENIS		&VIZIER\\
	&			&		&	&5.313	&4.490	&3.931	&	&	&2~MASS	&VIZIER\\
\par
23	&IRAS~10456		&		&	&3.45	&2.55	&2.15	&1.11	&	&SAAO		&T.~Lloyd Evans (mean)\\
	&			&		&	&3.354	&2.469	&2.045	&	&	&2~MASS	&VIZIER\\
\par
24	&HD~95767		&		&	&7.05	&6.48	&5.59	&3.50	&2.80	&ESO		&C.~Waelkens\\
	&			&		&8.786	&7.313	&	&4.964	&	&	&DENIS		&VIZIER\\
	&			&		&	&7.098	&6.532	&5.636	&	&	&2~MASS	&VIZIER\\
\par
25	&GK~Car		&43184.56	&	&9.05	&8.26	&8.14	&6.42	&	&SAAO		&\citet{85LloydEvans}\\
	&			&43159.52	&	&9.27	&8.46	&7.87	&6.64	&	&SAAO		&\citet{85LloydEvans}\\
	&			&		&10.150	&9.123	&	&7.970	&	&	&DENIS		&VIZIER\\
	&			&		&	&9.328	&8.716	&8.171	&	&	&2~MASS	&VIZIER	\\
\par
26	&IRAS~11472		&		&	&9.657	&9.047	&8.630	&	&	&2~MASS	&VIZIER\\
\par
27	&RU~Cen		&43188.51	&	&7.08	&6.65	&6.47	&6.53	&	&SAAO		&\citet{85LloydEvans}\\
	&			&43158.58	&	&7.14	&6.68	&6.39	&5.63	&	&SAAO		&\citet{85LloydEvans}\\
	&			&46243.25	&	&7.56	&7.04	&6.83	&6.33	&5.60	&SAAO		&\citet{87Goldsmith}\\
	&			&46248.28	&	&8.00	&7.44	&7.20	&	&	&SAAO		&\citet{87Goldsmith}\\
	&			&		&	&7.45	&7.02	&6.73	&6.08	&5.47	&ESO		&C.~Waelkens\\
	&			&		&	&7.68	&7.15	&6.86	&6.34	&5.80	&ESO		&\citet{97Garcia-Lario}\\
	&			&		&8.600	&7.151	&	&6.654	&	&	&DENIS		&VIZIER\\
	&			&		&	&7.616	&7.186	&6.918	&	&	&2~MASS	&VIZIER\\
\par
28	&SX~Cen		&43159.57	&	&7.94	&7.63	&6.88	&5.74	&	&SAAO		&\citet{85LloydEvans}\\
	&			&46243.26	&	&7.86	&7.25	&6.62	&5.20	&4.30	&SAAO		&\citet{87Goldsmith}\\
	&			&46248.30	&	&8.12	&7.51	&7.00	&5.65	&4.73	&SAAO		&\citet{87Goldsmith}\\
	&			&		&8.962	&7.660	&	&6.448	&	&	&DENIS		&VIZIER\\
	&			&		&	&7.875	&7.443	&6.759	&	&	&2~MASS	&VIZIER\\
\par
29	&HD~108015		&		&	&7.03	&6.39	&5.32	&3.50	&	&SAAO		&T.~Lloyd Evans (mean)\\
	&			&		&	&7.03	&6.40	&5.28	&3.14	&2.49	&ESO		&C.~Waelkens\\
	&			&		&	&6.941	&6.380	&5.338	&	&	&2~MASS	&VIZIER\\
30	&IRAS~13258		&50508.64	&	&11.65	&9.87	&8.29	&6.52	&	&SAAO		&T.~Lloyd Evans\\
	&			&50204.46	&	&12.11	&10.05	&8.48	&6.56	&	&SAAO		&T.~Lloyd Evans\\
	&			&		&14.474	&12.569	&	&	&	&	&DENIS		&VIZIER\\
	&			&		&	&12.160	&10.630	&9.303	&	&	&2~MASS	&VIZIER\\
31	&EN~TrA		&42945.38	&	&7.19	&6.67	&6.03	&4.69	&	&SAAO		&\citet{85LloydEvans}\\
	&			&		&	&7.10	&6.53	&5.73	&3.92	&3.38	&ESO		&C.~Waelkens\\
	&			&		&	&7.092	&6.578	&5.931	&	&	&2~MASS	&VIZIER	\\
32	&IRAS~15469		&		&	&7.24	&6.21	&4.89	&2.89	&	&SAAO		&T.~Lloyd Evans (mean)\\
	&			&		&	&7.190	&6.235	&4.967	&	&	&2~MASS	&VIZIER\\
33	&IRAS~15556		&48778.44	&	&9.42	&8.34	&6.89	&4.55	&	&SAAO		&T.~Lloyd Evans\\
	&			&50642.32	&	&9.51	&8.38	&6.88	&4.49	&	&SAAO		&T.~Lloyd Evans\\
	&			&		&10.877	&9.440	&	&6.887	&	&	&DENIS		&VIZIER\\
	&			&		&	&9.423	&8.350	&6.940	&	&	&2~MASS	&VIZIER\\
34	&IRAS~16230		&49124.51	&	&9.10	&8.48	&7.79	&6.29	&	&SAAO		&T.~Lloyd Evans\\
	&			&48819.36	&	&9.37	&8.77	&8.12	&6.26	&	&SAAO		&T.~Lloyd Evans\\
	&			&		&	&9.002	&8.510	&7.812	&	&	&2~MASS	&VIZIER\\
\par
35	&IRAS~17038		&50641.37	&	&7.74	&7.00	&6.58	&5.54	&	&SAAO		&T.~Lloyd Evans\\
	&			&49473.66	&	&8.36	&7.64	&7.16	&6.11	&	&SAAO		&T.~Lloyd Evans\\
	&			&		&9.434	&7.791	&	&6.369	&	&	&DENIS		&VIZIER\\
	&			&		&	&7.622	&7.050	&6.523	&	&	&2~MASS	&VIZIER\\
\par
36	&IRAS~17233		&50204.56	&	&10.03	&8.89	&7.28	&4.86	&	&SAAO		&T.~Lloyd Evans\\
	&			&49474.60	&	&10.89	&9.68	&7.72	&5.09	&	&SAAO		&T.~Lloyd Evans\\
	&			&		&	&10.423	&9.592	&8.371	&	&	&2~MASS	&VIZIER\\
\par
37	&IRAS~17243		&		&	&8.17	&7.46	&6.52	&4.75	&	&SAAO		&T.~Lloyd Evans (mean)\\
	&			&		&9.314	&8.177	&&6.592		&	&	&DENIS		&VIZIER\\
	&			&		&	&8.035	&7.358	&6.462	&	&	&2~MASS	&VIZIER\\
\par
38	&89~Her		&		&	&4.998	&4.239	&3.632	&	&	&2~MASS	&VIZIER\\
\par
39	&AI~Sco		&42997.35	&	&6.74	&5.94	&5.14	&3.79	&	&SAAO		&\citet{85LloydEvans}\\
	&			&46245.48	&	&6.62	&5.98	&5.33	&3.81	&	&SAAO		&\citet{87Goldsmith}\\
	&			&		&	&6.864	&6.179	&5.485	&	&	&2~MASS	&VIZIER\\
\par
40	&IRAS~18123		&		&	&8.04	&7.44	&6.86	&5.39	&	&SAAO		&T.~Lloyd Evans (mean)\\
	&			&		&	&7.974	&7.399	&6.729	&	&	&2~MASS	&VIZIER\\
\par
41	&IRAS~18158		&48778.52	&	&10.52	&9.31	&8.01	&6.40	&	&SAAO		&T.~Lloyd Evans\\
	&			&50204.60	&	&11.15	&9.80	&8.27	&6.17	&	&SAAO		&T.~Lloyd Evans\\
	&			&		&12.341	&10.958	&	&8.291	&	&	&DENIS		&VIZIER\\
	&			&		&	&10.788	&9.489	&8.120	&	&	&2~MASS	&VIZIER\\
\par
42	&AC~Her		&		&	&	&	&	&4.4	&4.0	&Kitt Peak NO	&\citet{70Gehrz}\\
	&			&		&	&	&	&	&	&	&Arizona system	&\\
	&			&		&	&	&	&5.2	&4.7	&3.6	&Mt Lemmon	&\citet{72bGehrz}\\
	&			&		&	&	&	&	&	&	&Arizona system	&\\
	&			&42456.87	&	&5.97	&5.52	&5.28	&4.73	&	&Tenerife	&\citet{85LloydEvans}\\
	&			&46243.52	&	&5.89	&5.39	&5.16	&4.66	&3.90	&SAAO		&\citet{87Goldsmith}\\
	&			&46248.47	&	&6.21	&5.66	&5.40	&4.87	&4.55	&SAAO		&\citet{87Goldsmith}\\
	&			&		&	&5.700	&5.338	&5.075	&	&	&2~MASS	&VIZIER\\
\par
43	&AD~Aql		&46253.53	&	&9.62	&9.14	&8.95	&	&	&SAAO		&\citet{87Goldsmith}\\
	&			&		&	&9.543	&9.134	&9.002	&	&	&2~MASS	&VIZIER\\
\par
44	&IRAS~19125		&		&	&7.48	&6.47	&5.38	&3.52	&	&SAAO		&T.~Lloyd Evans (mean)\\
	&			&		&	&8.356	&7.442	&7.067	&	&	&2~MASS	&VIZIER\\
45	&EP~Lyr		&		&	&	&	&	&6.6	&	&Mt Lemmon	&\citet{72bGehrz}\\
	&			&		&	&	&	&	&	&	&Arizona system	&\\
	&			&		&	&8.534	&8.153	&8.030	&	&	&2~MASS	&VIZIER	\\
46	&IRAS~19157		&48779.58	&	&8.89	&8.12	&6.95	&5.13	&	&SAAO		&T.~Lloyd Evans\\
	&			&48820.44	&	&8.94	&8.16	&6.92	&5.32	&	&SAAO		&T.~Lloyd Evans\\
	&			&		&9.713	&8.966	&	&7.109	&	&	&DENIS		&VIZIER\\
	&			&		&	&8.872	&8.205	&7.020	&	&	&2~MASS	&VIZIER\\
\par
47	&IRAS~20056		&45897.49	&	&9.99	&8.89	&7.31	&4.77	&	&SAAO		&\citet{88Menzies}\\
	&			&46693.29	&	&10.55	&9.20	&7.43	&4.78	&	&SAAO		&\citet{88Menzies}\\
	&			&		&	&10.097	&8.964	&7.471	&	&	&2~MASS	&VIZIER\\
\par
48	&R~Sge			&		&	&	&	&	&4.5	&3.3	&Kitt Peak NO	&\citet{70Gehrz}\\
	&			&		&	&	&	&	&	&	&Arizona system	&\\
	&			&		&	&	&	&	&4.6	&3.6	&Mt Lemmon	&\citet{72bGehrz}\\
	&			&		&	&	&	&	&	&	&Arizona system	&\\
	&			&46248.54	&	&7.10	&6.50	&5.80	&4.50	&	&SAAO		&\citet{87Goldsmith}\\
	&			&		&	&7.818	&7.234	&6.539	&	&	&2~MASS	&VIZIER	\\
\par
49	&V~Vul			&		&	&	&	&	&4.2	&3.4	&Mt Lemmon	&\citet{72bGehrz}\\
	&			&		&	&	&	&	&	&	&Arizona system	&\\
	&			&		&	&6.557	&6.019	&5.396	&	&	&2~MASS	&VIZIER\\
\par
50	&HD~213985		&		&	&8.18	&7.60	&6.68	&5.00	&4.45	&ESO		&C.~Waelkens\\
	&			&		&	&8.276	&7.609	&6.705	&	&	&2~MASS	&VIZIER\\
\par
51	&BD$+$39$^{\circ}$4926	&		&	&8.539	&8.394	&8.338	&	&	&2~MASS	&VIZIER\\\hline
\end{tabular}
\end{table}